Enhancing supply chain innovation and operational agility through knowledge acquisition from the social media: A microfoundational approach

Arsalan Mujahid Ghouri , Pervaiz Akhtar , V. G. Venkatesh , Aniqa Ashraf, Gayane Arsenyan, Shlomo Y. Tarba , and Zaheer Khan

Abstract

This paper presents an examination of the interlocks between knowledge acquisition from social media (KAfSM), organizational microfoundation structure and design (OMFSaD), supply chain innovation (SCI), and operational agility (OA). These interlocks were tested on data collected from 172 managers/directors/CEOs of 96 firms operating in nine manufacturing industry sectors in Malaysia. Our findings suggest that OMFSaD plays a key role when interlinked with KAfSM. Furthermore, OMFSaD is significantly associated with SCI and OA, and SCI significantly correlates with OA and partially mediates the relationship between OMFSaD and OA. Our study's outcomes are consistent with our understanding of IT-enabled organizational capabilities—thus contributing to dynamic capability theory—and suggest that KAfSM helps to revamp processes, routines, and business operations in frequently changing environments. In this paper, we draw implications for research and practice.

Keywords: Knowledge through social media, Microfoundations, Supply chain innovation, Operational agility, Dynamic capabilities, Manufacturing industry

Introduction

One of the best-known market and business axioms is that nothing is permanent except change. In this vein, the famous philosopher Heraclitus also proposed that change is the only reality. Businesses across the globe are continuously looking for ways to manage this 'permanent change' by developing new or leveraging existing capabilities suited to deal with globally preeminent challenges—e.g., volatility, uncertainty, complexity, and ambiguity. In such a context, operational agility (OA) is a vital capability whereby businesses can sense and respond to uncertain and dynamic business environments. In this regard, design, foresight, and systems thinking need to be supported in businesses by new skills and tools (i.e., [1], [2]). OA, which provides the strength and thinking needed to identify and seize any business opportunities, is considered a meta-capability of businesses (e.g., [3], [5]). Speed, accuracy, and cost-effectiveness—which the literature suggests to be integral parts of OA in relation to business success (e.g., [6], [8])—can be achieved by related knowledge, decision-making, networking, and communication. However, the extant literature has hitherto not investigated the benefits of organizational microfoundation structure and design (OMFSaD) and knowledge acquisition from social media (KAfSM), and the role that OA plays in such contexts. Moreover, to the best of our knowledge, no study on OMFSaD, KAfSM and the role played by OA has hitherto been conducted in the Malaysian manufacturing industry context.

OMFSaD refers to the underlying individual- and group-level actions that shape the strategically based partnerships with stakeholders, flexible decision-making, and open communication channels that lead to the emergence of superior organization-level performance (e.g., [9], [10]). [11] suggested that OA involves incremental and continuous change in an enterprise's core and adjacent business. Therefore, independence and skills - e.g., connecting with business partners, empowerment in decision making, and an environment in which all the

necessary information should be shared with all parties in order for them to complete their allocated tasks - could help manage the change and uncertainty present in the business environment. In our study, we addressed the relationship between OA, OMFSaD, and KAfSM in the presence of supply chain innovation (SCI). Some studies have investigated OA in relation to business organizational structure and design. For example, [12] conducted a case study on Haier— a Chinese multinational enterprise—and explained how an IT-based structure enhances the competency of a business in relation to processing information in turbulent business environments. In another IT-based study, [6] explored in depth the IT-enabled relationships among resources, structure, and routines and their effect on operational agility.

Further, [13] suggested that, in relation to stakeholder integration, customer connectivity, collaboration, and coordination, networking capability provides win-win solutions, knowledge creation and sharing, and the definition of new strategic directions. Also, flexible and adaptable behavioral characteristics are key enablers of business resilience, and flexible decision-making is meant to yield strategic crisis readiness (e.g., [14], [15]). Moreover, [8] shared a practical example of open communication. They discussed M.com's 'Big Ears' program, which stimulates open communication and collaboration across departments. This initiative connects line workers directly with senior management and acts as an internal digital channel suited to provide feedback on concerns or to propose innovations. Hence, previous research had hitherto not focused on how OMFSaD affects the extraction of KAfSM, and how it enhances SCI and OA in the context of manufacturing firms operating in emerging markets.

Malaysia is an emerging economy that is currently concentrating on major areas to add more value to business operations (i.e., [16], [17]). The Malaysian government is trying to concentrate on IT infrastructure to help the national industrial sectors (e.g., [18], [20]). Although the government of Malaysia has introduced a National Policy on Industry 4.0 (Industry4RWD) with the purpose of pushing manufacturing firms toward Industry 4.0, the related initiatives may not at this juncture be very visible among the general public, as the touch base is still rather confined to technocrats [21]. Therefore, research is needed to understand the depth of the initiatives taken by Malaysian manufacturing firms in regard to SCI and OA in the light of OMFSaD and KAfSM.

Digital technologies have transformed business activities and the linkages between all market and economy elements (i.e., 22], [23]), which has led to the evolution of digital supply networks (e.g., [4], [24], [25]). The social media, which have the strength to provide pivotal information in times of uncertainty in the business environment, could also provide scope to strengthen outside partnerships and networking, decision-making, and communication. For example, [7] and [8]—who conducted studies on the Internet of Things, dynamic data and information processing capabilities, and OA—found that resource interdependencies (pooled, sequential, and reciprocal) and IT-enabled capabilities (localized, synergistic, and optimized) are forged during operational processes to realize OA.

Innovation is an essential tool for businesses to deal with the dynamic changes in offerings and customer demands and problems (i.e., [26], [27]). Supply chain innovation (SCI) risk mitigation supports taking proactive and post-active measures aimed at supporting business operations (e.g., [28], [29]). Smooth and rich business organizational structure and design may facilitate supply chain innovation (e.g., [30], [31]).

By providing rich insights from firms based in an emerging economy, our study makes significant contributions to the literature by responding to the need for additional empirical research aimed at demonstrating the prominence of OMFSaD in enhancing OA. First, by drawing key insights from manufacturing firms based in an emerging economy, our study contributes to the OA literature—from the perspective of the manufacturing sector—by proposing a conceptual model of OA that involves KAfSM, OMFSaD, and SCI. Second, it contributes to the SCI literature by confirming the mediating role SCI plays between OMFSaD and OA in the manufacturing sectors of emerging markets. Its findings offer a holistic understanding of the chain of effects that leads to better OA in the context of an emerging economy. Third, our study contributes to the literature by highlighting the key role played by the KAfSM in OMFSaD in the manufacturing sector of an emerging economy. The related findings point to the role played by KAfSM in OMFSaD for manufacturing firm managers, which leads to the enhancement of OA in the manufacturing sector.

The remainder of this paper is structured as follows. Section 2 describes the relevant literature, theory, and underlying concepts. Section 3 presents the methodology and research design. Section 4 describes the empirical results and hypotheses testing. Finally, Section 5 provides conclusions, managerial and policy implications, and directions for future research.

Background Literature

Theoretical foundation

Our study is consistent with the dynamic capability theory (DCT) perspective. Dynamic capabilities contribute to the development of competitive advantages and, as such, they are of assistance in highly turbulent business environments (e.g., [32], [33]). The theory suggests that a business has a competitive advantage when it is able to effectively utilize its internal resources (e.g., employees and information technologies) to enhance its dynamic capability to quickly address changing business environments by sensing and seizing opportunities and accordingly transforming its capabilities [14]. Such capability suggests that social media and business structure

could provide the avenues and tools for businesses to create value and sustain themselves in dynamic and changing environments. To be precise, dynamic capabilities are those whereby businesses manage rapidly changing environments (e.g., [34], [35]). Previous studies have explained that the average period for a firm to sustain its competitive advantage has decreased over time (e.g., [36]-[38]).

The DCT perspective taken by previous studies is related to flexibility, supply chain integration, organizational learning, and knowledge management (e.g., [39]-[43]). [44] posited that dynamic capabilities are rooted in certain microfoundations. They further added that such capabilities enable the achievement of a congruence with technological opportunities suited to align with latent customer needs over time (doing the right things at the right time) by creating, extending, and/or revising ordinary capabilities and resource configurations. Our study built on the DCT literature to conceptualize and theoretically link social media competency, business organizational structure, and design strengths in achieving supply chain innovation and operational agility. The extensive use of social media prompts businesses to consider them as a tool suited to aid them in revamping their overall organizational structure and design to the end of motivating their employees to participate more in organizational routines and processes. [45] argued that the success of businesses relies on their knowledge of their external environments. For instance, businesses can extract from social media valuable knowledge suited to identify potential issues, problems, opportunities, and best practices [46]-[49]. The knowledge gleaned from social media contains information that has the potential to create value for a business [50]. External stakeholders hold valuable market information that could be hidden from businesses; therefore, OMFSaD could enhance SCI and provide steps and solutions suited to deal with disruptive environments. Therefore, businesses need to integrate their functions and processes with the available information in order to create dynamic capabilities with faster cycles and to ensure that all the data, insights, and intelligence gathered on changing market environments regularly feed their innovation processes in relation to managing dynamic environments (e.g., [1], [51], [52]).

Hypotheses Development

Knowledge acquisition from social media and organizational microfoundation structure and design

Social media is a source of information that covers all aspects of individual and business life. Social media has revolutionized how people share their knowledge, communicate, and collaborate while promptly engaging in conversations in the workplace [53], [54]. Businesses are increasingly and variously using social media tools for information sharing and communication (e.g., [55], [56]). Several organizations are utilizing social media to harness knowledge for value creation. For instance, IBM employs social media tools for knowledge sharing between its employees, thereby helping enhance their collaboration and innovation [57].

Furthermore, [58] posited that social media platforms are regarded as well-known and established venues of information-sharing channels that facilitate business organizational structure and design. Social media provides platforms and opportunities to connect with others who share the same interests and to exchange ideas with them. [59] described how social media fulfills several business functions—including cooperation, broadcast, dialogue, sociability, and stakeholder knowledge management. [60], [61] suggested that any knowledge acquired through social media helps to collaborate with others and facilitates a rapid flow of knowledge between people working across different geographical areas. Social media can change how a business evolves in a dynamic and uncertain economic era and facilitates the sharing of knowledge [62]-[64]. Additionally, social media hold promise as an inexpensive and relatively straightforward tool that can help an

organization become networked with its internal and external stakeholders, and, by so doing, increase its capacity to respond to the ills of society [65], [66]. Several studies have hinted that social media in the workplace enhance colleague relationships and work networks (e.g., [67],[69]).

Microfoundations is not a distinct theoretical or empirical approach per se; rather, it is a broad set of research heuristics and ways of problem-solving that call attention to lower and interlevel mechanisms in the fields of strategy, organization theory, micro organizational behavior, and applied psychology [70]. [71] hinted that microfoundations provide the idea that macro-level phenomena cannot be comprehended without an understanding of the individual-level factors that underpin them. Some studies have suggested that microfoundations pertain to the behaviors of agents in the firm—i.e., structure and design—and facilitate the enhancement or creation of a rigorous and consistent framework for macroeconomic analysis (e.g. [72], [73]). In our study's context, we posited that microfoundational structure and design motivate employees to participate more in organizational routines and processes, thus improving the sharing of knowledge to the end of enhancing supply chain innovation and operational agility.

Organizational structure and design enable decision-making on several levels. Any information available on specific matters helps to decide among the available choices. [74] proved that decision-making through the lens of information processing involves questions of how decisions emerge from basic cognitive processing—such as attention, memory, and causal reasoning. On the other hand, social media are rich in information and influence attention, memory, and causal reasoning. Therefore, social media can play a pivotal role in relation to decision-making (e.g., [75], [77]). [78] highlighted technological developments—e.g., social media—and suggested understanding how current and new technologies impact individuals, policy-making decisions, and the achievement of organizational objectives. [79] and [80]

recommended that social media in the workplace aid the sourcing of valuable information that can be used to help in decision-making processes. Social media offer continuous and diverse information; hence, they provide scope for flexible decision-making.

In relation to networking and flexible decision-making, social media provide new communication channels linking different stakeholders. This business communication structure enables internal employees to freely access and communicate both inside and outside the business environment. Many businesses encourage their employees to share their knowledge via social media because it facilitates efficient flows of information both within and between teams (e.g., [81], [83], [60]) explained that the benefits of open communication channels through social media enhance the provision of task information and capability, and the capacity to assist others, cooperate with other employees to develop ideas and solve problems or to implement policies and procedures. [84] exemplified that internal social media platforms can improve internal business operations and routines, information and knowledge sharing, and enhance internal open communication practice. [85] posited that open communication channels through social media create trust and diversity in businesses. We thus formulated the following hypothesis.

H1. Knowledge acquisition from social media is positively related to organizational microfoundation structure and design.

Organizational microfoundation structure and design, supply chain innovation, and operational agility

The ingredients of organizational structure and design are important aspects of any business's existence and progress. In regard to innovation, organizational structure and design are supportive pillars of value creation. [86] believed that discussing and thinking about innovation need to

include the recognition of an organization's structure, with its specific configurations and attitudinal orientations that set the business culture in relation to idea generation and risk taking.

The workforce's personal networks enable the input into the business of information and ideas that may be attained through formal or informal communication and observations. They provide the independence and space needed to think and deliberate freely with others on ideas and challenges. [87] explained that personal networks help create future offerings. [88] found that BetaRetail had solved specific issues through the systematic approach of their cross-functional sustainability team, which depended on people and their internal and external networks. [89] understood that the personal networks within a buying firm offer scope for supply chain innovations. These personal networks can focus on the problems and challenges of supply chains and yield solutions through the sharing of insightful thoughts. Additionally, they could help to identify and contact the entities best suited to find innovative solutions. It has been observed that these networks can also initiate inter-business collaboration aimed at developing innovative solutions to the pervasive challenges found in any operating ecosystem.

Data interchanges facilitate the flow of information along supply chains, resulting in knowledge and relational alignment [6], [90]. Flexibility in design gives rise to significant value creation and to the potential for radically changed economic risks [91], [92]. [93] and [94] argued that dynamic issues usually involve many factors and layers and are challenging to understand; therefore, flexible decision-making is a requisite to address those issues. [95] suggested that flexible decision-making styles with a better understanding of the situation are features of aggressive decision-makers with a high tolerance for ambiguity and uncertainty. Flexible decision-making supports different situations under complex structures and offers corresponding advice with structural adjustment (e.g., [96], [97]). Flexible decision-making is open and adaptable and

usually adjusted according to the business environment. Therefore, with such adjustments, businesses can attain relative advantages and compatibility in supply chains—i.e., innovation characteristics.

Further, [98] explained that businesses should maintain trust and open communication with their partners as essential ingredients of innovation. [99] elaborated that businesses should be able to use their innovation skills if they comply with their process requirements because employees can access and share information effortlessly and effectively through established rules, systems, processes, and procedures. [100] posited that SCI results from the improved business processes (as part of business organizational structure and design) required for effective supply chain management through collaborations between distributors, manufacturers, customers, and suppliers. Open communication facilitates collaboration between supply chain stakeholders, resulting in supply chain innovation. Additionally, an open communication environment empowers employees, which stimulates supply chain innovation. [101] and [102] discussed the role played by open communication in problem identification and in eliciting a sense of ownership of the supply chain process, which leads to problem-solving supply chain innovation. Based on this discussion, we proposed that:

H2. Organizational microfoundation structure and design positively relate to supply chain innovation.

OA focuses on quick responses, legitimate actions, and cost efficiency [7]. It is a capability whereby businesses sustain themselves and grow in their specific industry sectors, each in its specific environment and on the basis of different routines, processes, policies, and information flows. One of the components and characteristics of an agile business is large scale cooperation across several internal and external teams—i.e., personal networking (e.g., [103], [104]). [105]

explained that businesses may deal with agile environments through networked and empowered communities. Successful business executives and employees need to communicate effectively with both internal and external stakeholders (e.g., [106]). This improves communication and benefits businesses by enhancing relationships between them and their many stakeholders. Besides enabling businesses to share and adopt best practices, such relationship networks provide opportunities for supply chain innovation. [107] emphasized that production efficiency resources (i.e., structure and design) help to manage market dynamics and minimize environmental impact.

Speed and efficiency, as the ingredients of flexible decision-making, are considered critical components of lean and agile thinking. In agile business practice, employees are encouraged to communicate openly. Team transparency and openness and individual communication enable employees to quickly obtain the knowledge they need to make intelligent decisions [108]. Flexible decision-making procedures are iterative in making business practice evolve from generation to generation (e.g., [109], [110]. [111] and [112] hinted at establishing an environment that encourages interaction and mutual learning from experience to deal with agile conditions. A culture of open workplace communication may benefit employee morale and team cohesiveness [113]. Open communication provides employees with clarity in regard to their roles in highpressure and dynamic situations. [114] explained that information flows induce change and make it possible to respond to new challenges. [105] argued that comfortable information-sharing practices enable the exploration of new opportunities aimed at continuous improvement within the scope of wider shared goals. Therefore, open communication delivers up-to-date information and provides room for improvement and adequate decision-making in an industry's/market's current situation. Hence, we proposed the following hypothesis.

H3. Organizational microfoundation structure and design positively relate to operational agility.

Supply chain innovation and operational agility

SCI involves changes in supply chain networks, technologies, or processes (or combinations thereof) to enhance the creation of new value for stakeholders. Thus, SCI provides new ways to deal with daily routines, operations, processes, and decision-making. [115] explained that SCI is a process whereby a customer-oriented business adapts to environmental change. Businesses engage in SCI to pursue new ideas relating to products, services, processes, etc. This exploitation of new ideas pushes the boundaries of the capabilities of businesses. SCI could be about the adoption or creation of new software platforms and automation tools or the implementation of more sustainable and eco-friendly practices (i.e., [116], [118]). OA is a business's capability to improve its operations and processes to seize opportunities or sustain itself in the industry. OA covers an array of activities carried out by firms to create value in turbulent and unpredictable environments and enables them to identify opportunities early and mitigate any challenges presented by changes in external markets (e.g., [119], [121]). This requires flexible and adaptable supply chains capable of adjusting quickly to disruptions, shifts in demand, and other unexpected events. [107] and [122] illustrated that operational agility is an integral component of competitiveness, as it enables firms to swiftly and uninterruptedly adapt and react to any market fluctuations and helps organizations expand the variety of their offerings. Businesses need to embrace supply chain innovation and leverage cutting-edge technologies and practices in order to streamline their operations and improve their responsiveness in agile situations or environments.

Hence, we suggested the following hypothesis.

H4. Supply chain innovation is positively related to operational agility.

The literature hints at the fact that, by mediating between OMFSaD and OA, SCI may enhance the latter, and suggests that SCI is positively correlated with OMFSaD and OA. OMFSaD provides businesses with access to a plethora of information that can be useful to improve their SCI. Such information can help businesses to identify opportunities for SCI and enhance their OA. SCI can serve as a mediator between the OMFSaD and OA. By implementing innovative supply chain practices and social media technologies, businesses can improve their ability to respond quickly to changing market conditions. Thus, it can enhance the operational agility of businesses. Therefore, we proposed the following hypothesis.

H5: Supply chain innovation mediates the relationship between organizational microfoundation structure and design and operational agility.

Figure 1 depicts the model and hypotheses we developed based on the relevant literature.



Figure 1: Proposed conceptual framework

Methodology

Data collection and quality checks

Although the Malaysian government has been taking policy initiatives aimed at dealing with volatility, uncertainty, complexity, and ambiguity since 2019, local businesses had started considering it since 2018. The manufacturing sector is one of the most important ones in Malaysia; as stated by [123], it accounted for 22.3% of the country's GDP in 2019. That year, the Malaysian government introduced its National Inclusive and Sustainable Economic Policy (NISEP) to prepare the sector to face volatility, uncertainty, complexity, and ambiguity. A few local studies have hinted at the fact that Malaysian businesses had already started to think about facing uncertainty and unexpected challenges prior to 2019 and were ready to adapt their practices and routines as per a dynamic environment's requirements (i.e., [124], [125]). Thus, during the first half of 2022, we collected data from 172 managers/directors/CEOs of 96 firms operating in nine manufacturing industries connected with global value chains to understand their stance toward their respective industries' volatility, uncertainty, complexity, and ambiguity. To do so, we used an offline method that involved personal and colleagues' contacts, and personal visits to distribute a self-administered survey questionnaire that consisted of close-ended questions. We initially circulated 798 copies of our questionnaire (an average of 5.96 copies for each business). We received 185 responses—a 23.1% response rate. We retained 172 of these for the data analysis. We discarded the remaining 13 because they had either not been completed or more than one option had been chosen for one or more items. Our justification for collecting data at a single point of time was that it provided a snapshot of the Malaysian manufacturing context in the light of SCI and OA, as the aim of our study was to provide researchers with an understanding of the construct in its current form and to identify any patterns or trends that may be present in it. The other justification of our one-time data collection was to focus on specific aspects of OA at specific point in time. The demographic details of our final sample are presented in Table 1.

In order to maintain data quality, we checked for non-response bias by performing an independent t-test that involved comparing the first and last 20 respondents on all constructs (e.g., [126], [127]). The test results showed no significant difference between the early and late respondents. Therefore, the data was shown to be free from non-response bias. Furthermore, we also performed Harman's single-factor test to assess the potential for common method bias [128], which can cause systematic measurement errors that either inflate or deflate the observed relationships between constructs, generating both Type I and Type II errors (e.g., [129], [130]). The result of the variance explained by a single factor was 57.4, which implied the absence of any common method bias issues [131]. This showed us that our study's findings may be replicated in the future and enabled us to assume their generalizability (e.g., [132], [133]).

Table 1: Demographic details of respondents and businesses

Category		Frequency	%
Gender	Male	101	58.72
	Female	71	41.28
Designation	Manager	92	53.49
	Director	45	26.16
	CEO	35	20.35
	Never attended school	0	0
	Attended school	1	0.01
	Diploma	4	0.02
Education Level	Degree	33	19.19
	Masters	132	76.74
	PhD	2	0.01
	Less than 10 years	29	30.21
Business Age	10 years - 20 years	41	42.71
	More than 20 years	26	27.08
Industry Type	Food, Beverages & Tobacco	21	21.88
	Electronics	15	15.63
	Electrical	14	14.58
	Wood Furniture, Paper Products & Printing	12	12.5
	Transportation equipment	11	11.96
	Textiles, Wearing apparel, Leather & Footwear	8	8.33
	Motor vehicles	6	6.25
	Petroleum, Chemical, Rubber & Plastic	6	6.25
	Non-metallic minerals products, Basic metal & fabricated metal products	3	3.22

Operationalizing constructs

The KAfSM, OMFSaD, SCI, and OA constructs were obtained from existing studies. [10] defined the three items we adopted for OMFSaD, and [134], [136] identified the five for KAfSM. [137], [139] helped to finalize the five items for SCI, and [7] the eight for OA (refer to Appendix A for the questionnaire).

Control variables

We added business age (BA) and Industry Type (InT) as control variables for our study.

Some previous studies have used BA as a control variable (e.g., [141], [143]), as businesses of

different ages may have different capabilities to deal with uncertain environments. [145] hinted at the fact that young businesses usually have less experience from which to learn. [146] also found that younger and older businesses may vary in how and what they learn. These variations in learning capacity impact the promptness of businesses' responses to the speed and rate of the changes they face. [146] further added that younger businesses are more agile and 'fit' for change (e.g., reconfiguring their routines) than older ones.

InT has also been adopted as a control variable (e.g., [147], [149]) in the literature. [150] showed that industry type moderates the effects of public concern, regulatory forces, top management commitment, and competitive advantage on environmental orientation and strategy. Businesses operating in different industries face unique external environmental conditions and internal contingencies (e.g., technological ones) [151]. [151] provided further evidence that a business's success is related to its ability to select a business-level strategy that maximizes the available resources in a specific industry setting. An industry's characteristics thus affect the nature of the competitive actions undertaken by businesses within it.

For the BA variable, we assigned values ranging from 1 to 3 to represent business ages of less than 10 years, between 10 and 20 years, and more than 20 years, in ascending order. For the InT variable, we assigned values from 1 to 9 denote the 'food, beverages & tobacco', 'electronics', 'electrical', 'wood furniture, paper products & printing', 'transportation equipment', 'textiles, wearing apparel, leather & footwear', 'motor vehicles', 'petroleum, chemical, rubber & plastic', and 'non-metallic minerals products, primary metal & fabricated metal products' sectors, in order. As advocated by [140], we justified the addition of control variables to our study as they would enable us to isolate any selection bias in a specific observation group and to control for variations

in R². Table 2 shows the details of the constructs, each of which was measured on a five-point Likert scale.

Analysis and Results

Measurement model

We obtained our results through confirmatory factor analysis (CFA), and tested our hypotheses through mediation analysis. As a statistical approach suited to validate how well a proposed model matches the data, we performed CFA to determine the hypothesized model's validity, reliability, and fitness. Compared to its analytical counterpart—i.e., exploratory factor analysis—CFA confirms the relationship between the observed and the underlying latent factors for the instruments already tested (e.g., [152]). CFA belongs to the broader family of methods known as Structural Equation Modeling (SEM) and plays a vital role in validating and evaluating measurement models [153].

One of the prominent features of CFA is its ability to reliably and accurately evaluate highly complex theories. These theories are first transformed into models and then tested statistically. Along these lines, [154] suggested that the model to be assessed should be explicitly defined, estimated, and validated. Then, model fitness can be evaluated through derived model estimates. Apart from chi-square fitness scores, several global fitness indices have been proposed, such as the *Tucker-Lewis (TLI), comparative fit (CFI), and root mean square error of approximation (RMSEA) indices* [155]. The following section reports the model fit indices results, which are presented in Table 2.

As per measurement model results, the CMIN/DF is the chi-square value divided by the degrees of freedom—also known as relative chi-square. CMIN/DF values lower than 3 indicate a good model fit. The value for the current model was found to be 1.533, thus confirming model fit.

The TLI and the CFI values were found to be 0.962 and 0.974, respectively. As they fell above the minimum threshold value of 0.95, they were found to support an acceptable range of model fit. Finally, the RMSEA value (0.049, lower than the 0.07 maximum threshold) was found to also suggest that the model was a good fit. To summarize, the hypothesized model was a good fit and thus suitable for further analysis.

Table 2 further demonstrates the reliability and validity of each of our four constructs i.e., IDE, KAfSM, SCI, and OA. We determined convergent validity through factor loadings and AVE values. According to [156], factor loadings are good if 0.70 or higher, acceptable between 0.5 and 0.7, and unacceptable if lower than 0.5. Additionally, convergent validity is indicated by a construct's average variance extracted (AVE) value higher than 0.5, as specified by [157]. In our study, the factor loadings and AVE values of all the individual factors were found to satisfy those criteria, confirming acceptable convergent validity.

In line with the above, the reliability statistics for the IDE, KAfSM, SCI, and OA constructs were found to have Cronbach's Alpha coefficient values of 0.799, 0.817, 0.889, and 0.767, respectively. This was a measure of the good reliability of the items used in the instrument, as all those values were more significant than the minimum threshold of 0.7 [158]. Consequently, the hypothesized model was found to be free of reliability and convergent validity issues.

Construct	Item Code	Loadings	Cronbach's Alpha	Average Variance Extracted	
KAfSM	KAfSM_3	0.88		0.631	
	KAfSM_4	0.79			
	KAfSM_4	0.76	0.831		
	KAfSM_5	0.71			
	KAfSM_1	0.66			
	MFO_2	0.92		0.561	
OMFSaD	MFO_1	0.82	0.843		
	MFO _3	0.78			
	SCI_1	0.89		0.727	
	SCI_2	0.72			
SCI	SCI_4	0.66	0.772		
	SCI_5	0.65			
	SCI_3	0.62			
OA	OA_2	0.78		0.536	
	OA_5	0.88			
	OA_1	0.81			
	OA_7	0.77	0 799		
	OA_6	0.71	0.788		
	OA_8	0.68			
	OA_3	0.64			
	OA_4	0.60			

Table 2: Construct details, loading, reliability, and convergent validity

Note: KAfSM = Knowledge Acquisition from Social Media, OMFSaD = Organization's microfoundation structure and design, SCI = Supply Chain Innovation, OA = Operational Agility

Regarding discriminant validity, the approach proposed by [157] suggests that the correlation value must be lower than the AVE values between the constructs. In order to confirm discriminant validity, the square root of AVE must be higher than the correlations between the factors [159]. Table 3 shows the appropriateness of discriminant validity as each construct was found to account for more variance in its own associated indicator variable than in those shared with other constructs.

Table 3: Discriminant validity

Constructs	KAfSM	OMFSaD	SCI	OA
KAfSM	0.788			
OMFSaD	0.211	0.772		
SCI	0.092	0.122	0.866	
OA	0.357	0.389	0.091	0.754

Note: KAfSM = Knowledge Acquisition from Social Media, OMFSaD = organization's microfoundation structure and design, SCI = Supply Chain Innovation, OA = Operational Agility

Hypotheses testing

We formulated and tested five hypotheses. The detailed results of the path and mediation analyses are presented below in Table 4.

The first hypothesis was related to the direct impact of KAfSM on OMFSaD. The estimated coefficient between KAfSM and OMFSaD was found to be 0.246 and significant (p > 0.01). According to [160], if the calculated critical value is higher than 1.96 (i.e., p > 0.05), a hypothesis will be considered supported. Therefore, the impact of KAfSM on OMFSaD was found to be significant and positive, and H1 was supported. KAfSM thus has an affirmative relationship with OMFSaD in the Malaysian manufacturing industry context.

The second hypothesis considered the direct impact of OMFSaD on SCI. The estimated coefficient between OMFSaD and SCI was found to be 0.398 and significant (p > 0.01). Therefore, the impact of OMFSaD on SCI was found to be significant and positive, and H2 was supported. OMFSaD thus has an assenting relationship with SCI in the Malaysian manufacturing industry context.

The third hypothesis pertained to the direct impact of OMFSaD on OA. The estimated coefficient between OMFSaD and OA was found to be 0.429 and significant (p > 0.05). Therefore, the impact of OMFSaD on OA was found to be significant and positive, and H3 was supported.

OMFSaD thus has a positive relationship with OA in the Malaysian manufacturing industry context.

The fourth hypothesis focused on the direct impact of SCI on OA. The estimated value between SCI and OA was found to be 0.338 and significant (p > 0.01). Therefore, the impact of SCI on OA was found to be significant and positive, and H4 was supported. SCI thus has an affirmative relationship with OA in the Malaysian manufacturing industry context.

Hypothesis 5 reflected the mediating effect of SCI between OMFSaD and OA. The critical value of the link was found to decrease but to still be significant, and thus to partially mediate. SCI was thus found to partially explain the relationship of OMFSaD and OA in the Malaysian manufacturing industry context.

Paths	Estimate	S.E.	C.R.	Hypotheses
Direct Effects				
KAfSM => OMFSaD	0.246	0.145	3.388**	H1
OMFSaD => SCI	0.398	0.267	7.117***	H2
OMFSaD => OA	0.255	0.037	4.002***	H3
SCI => OA	0.338	0.061	6.653***	H4
Indirect Effect				
OMFSaD => SCI=> OA	0.154	0.041	3.307**	H5
	KAfSM	OMFSaD	SCI	
R^2	0.224**	0.1455**	0.3117**	

Table 4: Path results of the model

Note: statistically significant at *** (p < 0.01), and ** (p < 0.05)

Discussion and Implications

The main aim of our study was to understand the role played by the acquisition of knowledge from social media on organizational microfoundation structure and design and that of the latter on

operational agility (with the mediating effect of supply chain innovation) in the context of emerging markets' manufacturing firms connected with global value chains. Our study provided an insight into how firms achieve operational agility through innovation and firm structure and design, and by harnessing knowledge from social media. Our findings contributes to the literature in the following manner. First, we checked for the impact of knowledge acquisition from social media on organizational microfoundation structure and design in key sectors of the manufacturing industry and found that a rich knowledge acquisition from social media strengthens organizational microfoundation structure and design. Second, we tested for the impact of organizational microfoundation structure and design on operational agility. We also checked for the role played by supply chain innovation, confirming that it is integrated between organizational microfoundation structure and design and operational agility. Several studies have concluded that social media enhance business capabilities and provide a competitive advantage [161, 162]. Conversely, organizational microfoundation structure and design had not hitherto been checked with operational agility in the manufacturing sector in the presence of supply chain innovation, specifically in an emerging market context. Organizational microfoundation structure and design provide the bases and strength to prepare a business to face an uncertain environment. The manufacturing sector literature does not clearly show the benefits of organizational microfoundation structure and design in regard to enhancing operational agility capability. Our results support the theoretical framework and related hypotheses and confirm the relationship between organizational microfoundation structure and design, supply chain innovation, and operational agility in emerging markets. Our study provides an understanding of the importance of organizational microfoundation structure and design for businesses and of the extraction of knowledge from social media in building/sustaining their capabilities (i.e., operational agility). These findings offer important implications.

Theoretical contributions

Our study is aligned with the DCT perspective that technology and firm infrastructure provide grounds to develop competitive advantages by enhancing a firm's dynamic capability. Our findings imply that social media avenues and resources enable businesses to build value across intermediate organizational capabilities—e.g., operational agility (e.g., [39], [163], [164]). Recently, several studies have taken the DCT perspective to examine the business organizational structure and design capabilities integration with social media (e.g., [165], [166]).

Our study provides the following theoretical implications. The first is that organizational microfoundation structure and design enable emerging market firms to develop capabilities that enhance the forecasting and preparatory skills they need to deal with volatile, uncertain, complex, and ambiguous environments. The acquisition of knowledge from social media offers opportunities to streamline and improve organizational microfoundation structure and design, which, in turn, enhances innovation and operational agility. The acquisition of knowledge from social media favors the prospects of strengthening organizational microfoundation structure and design.

The second implication is linked to our finding that the information extracted from organizational microfoundation structure and design helps revamp business processes, routines, and operations in challenging situations through the acquisition of vital knowledge from social media [161]. Besides, organizational microfoundation structure and design can add value to customer offerings, as our results indicate that organizational microfoundation structure and design are positively linked with supply chain innovation, structure, and design in relation to proposing

new ideas and inputs to be shared by internal stakeholders, and subsequently adds value to the supply chain processes.

The third implication is that organizational microfoundation structure and design help businesses deal with their competitors, predict and respond to market changes, bring fast and flexible offerings, and quickly respond to marketing opportunities. Businesses endowed with effective organizational microfoundation structure and design can identify any potential issues, problems, opportunities, and best solutions and practices, as per their industry's requirements (e.g., [167]).

Practical implications

Our study provides business managers and policymakers with several implications and further insights into operational agility. The first implication for managers is that acquiring knowledge from social media can assist in organizational microfoundation structure and design, a perspective that turns personal networks, flexible decision-making, and open communication into business assets. The results of our study provide managers with empirical evidence that knowledge sourced from social media can aid businesses in achieving effective structure and design which can be conducive to enhancing their innovation and operational agility. At the same time, our results show policymakers that relevant knowledge sourced from social media enables businesses to identify opportunities to improve. Therefore, policymakers should work to establish an IT-enabled environment across society at large, so that employees can benefit from it and help to enhance their existing organizational microfoundation structure and design.

The second implication is that organizational microfoundation structure and design enhance supply chain innovation. [42] demonstrated that organizational microfoundation structure and design facilitate supply chain innovation. Our study's findings advise managers to engage both themselves and employees in the use of personal networks, encourage flexible decision-making when needed, and facilitate open communication in order to deal with the marketplace and business environment and achieve supply chain innovation. Based on our study's findings, organizational microfoundation structure and design should cater to the attainment of competitive differential advantages, like differentiating processes or products. These differentiations could be based on the information and material flow, intra- and inter-business coordination, and collaboration.

Furthermore, organizational microfoundation structure and design are pivotal for managers in their management of specific product lifecycle phases, which is an integral part of supply chain innovation. Before or after the launch of a new product, managers seek different types of information (e.g., [168], [170]). [171] believed that supply chain innovation helps to launch new products more effectively. In the growth phase, managers seek employee involvement in product enhancement or management, which is crucial to improve a company's product offerings. In the maturity phase, managers need the information to increase the appeal of their products, and thus maintain/increase market share. Additionally, any market and competitor information gathered internally is more crucial in the maturity phase. Finally, during the decline phase, managers would become aware internally (through their personal networks) that a product needs to be replaced with a newer or related offering.

The third implication is that organizational microfoundation structure and design enhance operational agility. This also highlights the effect of supply chain innovation as a mediator. Managers usually operate according to specific business and operational models; therefore, we posit that, in the presence of supply chain innovation, the inputs gained from organizational microfoundation structure and design help transform processes and operations in uncertain environments. Managers should pay careful attention to the criticisms and observations expressed by employees. In the current environment, customers evolve faster than businesses (e.g., [172]), and employees can share or extract the related information from the outside world. Hence, managers need to gain the ability to overcome any deficiencies in competitive environments. Personal networks, flexible decision-making, and open communication enable B2C businesses to achieve an understanding of their customers. This understanding, which enlists the help of organizational microfoundation structure and design (in the presence of supply chain innovation), makes room for the process and operational flexibility needed to adjust to different goals, as per the environment's requirements. At the same time, organizational microfoundation structure and design (in the presence of supply chain innovation) afford managers and businesses the responsiveness they need regarding market requirements. Last but not least, organizational microfoundation structure and design (in the presence of supply chain innovation) provide businesses with the robustness they need to develop the ability to absorb or address the effects of environmental uncertainty. In uncertain situations, businesses are deemed to need support that varies with age, experience, and type of business. Therefore, policymakers should enact policies or frameworks suited to address the possible issues and solutions that may present themselves in certain situations.

Conclusion

Operational agility is considered one of the capabilities vital to sense and respond to dynamic and unseen challenges. Our study provides evidence that organizational microfoundation structure and design play a part in helping businesses to attain a competitive advantage through the acquisition of knowledge. We thus proposed five hypotheses and tested them on a sample of 172 managers/directors/ CEOs from 96 firms operating in nine manufacturing industries in Malaysia—

one of the most important emerging markets. All five hypotheses were found to be supported. [6] stressed the importance of operational agility and provided evidence for the role played in its improvement.

Limitations and future directions

With its key contributions to the literature and its implications for theory and practice, our study opens up several future research avenues associated with its limitations in terms of methodology and boundaries. First, we used data collected from nine prominent manufacturing sectors in Malaysia. Hence, future studies could test our proposed framework in other geographical contexts and other industries of the manufacturing sector. This approach could involve comparative studies between sectors and economies (i.e., developed, emerging, and transitional ones.). Second, future studies could include other potential mediating (e.g., sustainable supply chain) and moderating (e.g., institutions) variables. Third, a qualitative or mixed approach (with a proposed framework) could also be an option suited to extend our study's findings. Fourth, future studies could target customers to perform an in-depth analysis aimed at understanding why and under which circumstances they share vital information with businesses on social media. Customer behavior tends to vary under dynamic circumstances; examining the role of customers in enhancing supply chain innovation operational agility could thus provide important insights. Fifth, collecting data on a similar model at two different points in time could provide the opportunity to understand how knowledge acquisition from social media evolves and leads to innovation and operational agility. Dynamic capabilities-e.g., operational agility-vary quickly in market and industry environments; future studies could thus examine different sets of dynamic capabilities and microfoundations and their role in supply chain innovation and operational agility across developed and emerging markets.

References:

 C. C. J. M. Millar, O. Groth, and J. F. Mahon, "Management Innovation in a VUCA World: Challenges and Recommendations," Calif Manage Rev, vol. 61, no. 1, pp. 5–14, Nov. 2018, doi: 10.1177/0008125618805111.

[2] M. Marinković, O. Al-Tabbaa, Z. Khan, and J. Wu, "Corporate foresight: A systematic literature review and future research trajectories," J Bus Res, vol. 144, pp. 289–311, May 2022, doi: 10.1016/j.jbusres.2022.01.097.

[3] Y. Doz and M. Kosonen, "The Dynamics of Strategic Agility: Nokia's Rollercoaster Experience," Calif Manage Rev, vol. 50, no. 3, pp. 95–118, Apr. 2008, doi: 10.2307/41166447.

[4] P. Akhtar, A. M. Ghouri, M. Saha, M. R. Khan, S. Shamim, and K. Nallaluthan, "Industrial Digitization, the Use of Real-Time Information, and Operational Agility: Digital and Information Perspectives for Supply Chain Resilience," IEEE Trans Eng Manag, pp. 1–11, 2022, doi: 10.1109/TEM.2022.3182479.

[5] D. E. Strode, H. Sharp, L. Barroca, P. Gregory, and K. Taylor, "Tensions in Organizations Transforming to Agility," IEEE Trans Eng Manag, vol. 69, no. 6, pp. 3572–3583, Dec. 2022, doi: 10.1109/TEM.2022.3160415.

[6] F. T. C. Tan, B. Tan, W. Wang, and D. Sedera, "IT-enabled operational agility: An interdependencies perspective," Information & Management, vol. 54, no. 3, pp. 292–303, Apr. 2017, doi: 10.1016/j.im.2016.08.001.

[7] P. Akhtar, Z. Khan, S. Tarba, and U. Jayawickrama, "The Internet of Things, dynamic data and information processing capabilities, and operational agility," Technol Forecast Soc Change, vol. 136, pp. 307–316, Nov. 2018, doi: 10.1016/j.techfore.2017.04.023.

[8] F. T. C. Tan, S. L. Pan, and M. Zuo, "Realising platform operational agility through information technology–enabled capabilities: A resource-interdependence perspective," Information Systems Journal, vol. 29, no. 3, pp. 582–608, May 2019, doi: 10.1111/isj.12221.

U. Elg, P. N. Ghauri, J. Child, and S. Collinson, "MNE microfoundations and routines for building a legitimate and sustainable position in emerging markets," J Organ Behav, vol. 38, no. 9, pp. 1320–1337, Nov. 2017, doi: 10.1002/job.2214.

[10] A. M. Ghouri, P. Akhtar, M. Shahbaz, and H. Shabbir, "Affective organizational commitment in global strategic partnerships: The role of individual-level microfoundations and social change," Technol Forecast Soc Change, vol. 146, pp. 320–330, Sep. 2019, doi: 10.1016/j.techfore.2019.05.025.

[11] S. Denning, The age of agile: how smart companies are transforming the way work gets done. New York: American Management Association, 2018.

[12] P.-Y. Huang, T. H. Ouyang, S. L. Pan, and T.-C. Chou, "The role of IT in achieving operational agility: A case study of Haier, China," Int J Inf Manage, vol. 32, no. 3, pp. 294–298, Jun. 2012, doi: 10.1016/j.ijinfomgt.2012.02.001.

[13] C. Battistella, A. F. de Toni, G. de Zan, and E. Pessot, "Cultivating business model agility through focused capabilities: A multiple case study," J Bus Res, vol. 73, pp. 65–82, Apr. 2017, doi: 10.1016/j.jbusres.2016.12.007.

[14] J. Vargo and E. Seville, "Crisis strategic planning for SMEs: finding the silver lining," IntJ Prod Res, vol. 49, no. 18, pp. 5619–5635, Sep. 2011, doi: 10.1080/00207543.2011.563902.

[15] R. Pal, H. Torstensson, and H. Mattila, "Antecedents of organizational resilience in economic crises—an empirical study of Swedish textile and clothing SMEs," Int J Prod Econ, vol. 147, pp. 410–428, Jan. 2014, doi: 10.1016/j.ijpe.2013.02.031.

[16] Ghouri, A. M., Akhtar, P., Haq, M. A., Mani, V., Arsenyan, G., & Meyer, M. (2021).Real-time information sharing, customer orientation, and the exploration of intra-service industry differences: Malaysia as an emerging market. Technol Forecast Soc Change, 167, 120684.

[17] Ooi, S. M., & Richardson, C. (2019). The internationalisation of service-sector SMEs in an emerging market: Insights from business training and consultancy firms in Malaysia. Review of International Business and Strategy, 29(1), 44-60.

[18] Malik, S. A., & Ayop, A. R. (2020). Solar energy technology: Knowledge, awareness, and acceptance of B40 households in one district of Malaysia towards government initiatives. Technology in Society, 63, 101416.

[19] Lai, S. T. (2023). FMM: Budget 2023 lacks specific support for manufacturing. DNA.Retrieved from

https://www.digitalnewsasia.com/business/fmm-budget-2023-lacks-specific-supportmanufacturing

[20] PR Newswire (2023). The logistics industry in Malaysia is expected to contribute more than \$350 Bn by 2027F owing to the government initiatives to improve cross border infrastructure and Malaysia's growing FTAs: Ken Research. Yahoo Finance. Retrieved from <u>https://finance.yahoo.com/news/logistics-industry-malaysia-expected-contribute-</u>

<u>112800310.html</u>

[21] Ling, Y. M., Hamid, N. A. A., & Te Chuan, L. (2020). Is Malaysia ready for Industry 4.0?Issues and challenges in manufacturing industry. International Journal of Integrated Engineering, 12(7), 134-150.

[22] M. Hannibal and G. Knight, "Additive manufacturing and the global factory: Disruptive technologies and the location of international business," International Business Review, vol. 27, no. 6, pp. 1116–1127, Dec. 2018, doi: 10.1016/j.ibusrev.2018.04.003.

 [23] J. Zeng, Z. Khan, and M. de Silva, "The emergence of multi-sided platform MNEs: Internalization theory and networks," International Business Review, vol. 28, no. 6, p. 101598, Dec. 2019, doi: 10.1016/j.ibusrev.2019.101598.

[24] J. Wareham, L. Mathiassen, A. Rai, D. Straub, and R. Klein, "The business value of digital supply networks: A program of research on the impacts of globalization," Journal of International Management, vol. 11, no. 2, pp. 201–227, Jun. 2005, doi:

10.1016/j.intman.2005.03.007.

[25] M. Du, Q. Chen, J. Xiao, H. Yang, and X. Ma, "Supply Chain Finance Innovation Using Blockchain," IEEE Trans Eng Manag, vol. 67, no. 4, pp. 1045–1058, Nov. 2020, doi: 10.1109/TEM.2020.2971858.

[26] M. U. Ahmed and A. Shafiq, "Toward sustainable supply chains: impact of buyer's legitimacy, power and aligned focus on supplier sustainability performance," International Journal of Operations & Production Management, vol. 42, no. 3, pp. 280–303, Mar. 2022, doi: 10.1108/IJOPM-08-2021-0540.

[27] S. Solaimani and J. van der Veen, "Open supply chain innovation: an extended view on supply chain collaboration," Supply Chain Management: An International Journal, vol. 27, no. 5, pp. 597–610, Jul. 2022, doi: 10.1108/SCM-09-2020-0433.

[28] M. F. Afraz, S. H. Bhatti, A. Ferraris, and J. Couturier, "The impact of supply chain innovation on competitive advantage in the construction industry: Evidence from a moderated

multi-mediation model," Technol Forecast Soc Change, vol. 162, p. 120370, Jan. 2021, doi: 10.1016/j.techfore.2020.120370.

[29] W. Liu, J. Zhang, and S. Wang, "Factors influencing the smart supply chain innovation performance of commodity distribution enterprises: an investigation from China," Industrial Management & Data Systems, vol. 121, no. 10, pp. 2073–2099, Oct. 2021, doi: 10.1108/IMDS-12-2020-0753.

[30] K. B. Kahn, "Understanding innovation," Bus Horiz, vol. 61, no. 3, pp. 453–460, May 2018, doi: 10.1016/j.bushor.2018.01.011.

[31] X. Chen, C. Wang, and S. Li, "The impact of supply chain finance on corporate social responsibility and creating shared value: a case from the emerging economy," Supply Chain Management: An International Journal, Feb. 2022, doi: 10.1108/SCM-10-2021-0478.

[32] D. J. Teece, "Explicating dynamic capabilities: the nature and microfoundations of
(sustainable) enterprise performance," Strategic Management Journal, vol. 28, no. 13, pp. 1319–
1350, Dec. 2007, doi: 10.1002/smj.640.

[33] D. Teece, M. Peteraf, and S. Leih, "Dynamic Capabilities and Organizational Agility:
 Risk, Uncertainty, and Strategy in the Innovation Economy," Calif Manage Rev, vol. 58, no. 4, pp. 13–35, Aug. 2016, doi: 10.1525/cmr.2016.58.4.13.

[34] S. Ghosh, M. Hughes, I. Hodgkinson, and P. Hughes, "Digital transformation of industrial businesses: A dynamic capability approach," Technovation, vol. 113, p. 102414, May 2022, doi: 10.1016/j.technovation.2021.102414.

[35] E. Ramos, A. S. Patrucco, and M. Chavez, "Dynamic capabilities in the 'new normal': a study of organizational flexibility, integration and agility in the Peruvian coffee supply chain,"

Supply Chain Management: An International Journal, Oct. 2021, doi: 10.1108/SCM-12-2020-0620.

[36] Barreto, I. (2010). Dynamic capabilities: A review of past research and an agenda for the future. Journal of Management, 36(1), 256-280.

[37] Schilke, O., Hu, S., & Helfat, C. E. (2018). Quo vadis, dynamic capabilities? A contentanalytic review of the current state of knowledge and recommendations for future research. Academy of Management Annals, 12(1), 390-439.

[38] Wang, C. L., & Ahmed, P. K. (2007). Dynamic capabilities: A review and research agenda. International Journal of Management Reviews, 9(1), 31-51.

[39] I. Ali, I. Golgeci, and A. Arslan, "Achieving resilience through knowledge management practices and risk management culture in agri-food supply chains," Supply Chain Management: An International Journal, Dec. 2021, doi: 10.1108/SCM-02-2021-0059.

[40] S. Bag and M. S. Rahman, "The role of capabilities in shaping sustainable supply chain flexibility and enhancing circular economy-target performance: an empirical study," Supply Chain Management: An International Journal, Oct. 2021, doi: 10.1108/SCM-05-2021-0246.

[41] H. Do, P. Budhwar, H. Shipton, H.-D. Nguyen, and B. Nguyen, "Building organizational resilience, innovation through resource-based management initiatives, organizational learning and environmental dynamism," J Bus Res, vol. 141, pp. 808–821, Mar. 2022, doi:

10.1016/j.jbusres.2021.11.090.

[42] A. Liu, H. Liu, and J. Gu, "Linking business model design and operational performance:
The mediating role of supply chain integration," Industrial Marketing Management, vol. 96, pp. 60–70, Jul. 2021, doi: 10.1016/j.indmarman.2021.04.009.

[43] Wilden, R., Gudergan, S. P., Nielsen, B. B., & Lings, I. (2013). Dynamic capabilities and performance: strategy, structure and environment. Long Range Planning, 46(1-2), 72-96.
[44] Magistretti, S., Ardito, L., & Messeni Petruzzelli, A. (2021). Framing the microfoundations of design thinking as a dynamic capability for innovation: Reconciling theory and practice. Journal of Product Innovation Management, 38(6), 645-667.

[45] B. Nguyen, X. Yu, T. C. Melewar, and J. Chen, "Brand innovation and social media: Knowledge acquisition from social media, market orientation, and the moderating role of social media strategic capability," Industrial Marketing Management, vol. 51, pp. 11–25, Nov. 2015, doi: 10.1016/j.indmarman.2015.04.017.

[46] P. del Vecchio, G. Mele, G. Passiante, D. Vrontis, and C. Fanuli, "Detecting customers knowledge from social media big data: toward an integrated methodological framework based on netnography and business analytics," Journal of Knowledge Management, vol. 24, no. 4, pp. 799–821, May 2020, doi: 10.1108/JKM-11-2019-0637.

[47] M. Salehan and D. J. Kim, "Predicting the performance of online consumer reviews: A sentiment mining approach to big data analytics," Decis Support Syst, vol. 81, pp. 30–40, Jan. 2016, doi: 10.1016/j.dss.2015.10.006.

[48] Jafari-Sadeghi, V., Mahdiraji, H. A., Busso, D., & Yahiaoui, D. (2022). Towards agility in international high-tech SMEs: Exploring key drivers and main outcomes of dynamic capabilities. Technol Forecast Soc Change, 174, 121272.

[49] Laaksonen, O., & Peltoniemi, M. (2018). The essence of dynamic capabilities and their measurement. International Journal of Management Reviews, 20(2), 184-205.

[50] G. Tomas and M. Hult, "An integration of thoughts on knowledge management,"Decision Sciences, vol. 34, no. 2, pp. 189–195, May 2003, doi: 10.1111/1540-5915.02264.

[51] C. Millar, M. Lockett, and T. Ladd, "Disruption: Technology, innovation and society,"Technol Forecast Soc Change, vol. 129, pp. 254–260, Apr. 2018, doi:

10.1016/j.techfore.2017.10.020.

[52] S. K. Vickery, X. Koufteros, and C. Droge, "Does product platform strategy mediate the effects of supply chain integration on performance? A dynamic capabilities perspective," IEEE Trans Eng Manag, vol. 60, no. 4, pp. 750–762, Nov. 2013, doi: 10.1109/TEM.2013.2266301.

[53] S. Kwayu, M. Abubakre, and B. Lal, "The influence of informal social media practices on knowledge sharing and work processes within organizations," Int J Inf Manage, vol. 58, p. 102280, Jun. 2021, doi: 10.1016/j.ijinfomgt.2020.102280.

[54] H. Li and Y. Sakamoto, "Social impacts in social media: An examination of perceived truthfulness and sharing of information," Comput Human Behav, vol. 41, pp. 278–287, Dec. 2014, doi: 10.1016/j.chb.2014.08.009.

[55] S. Cartwright, I. Davies, and C. Archer-Brown, "Managing relationships on social media in business-to-business organisations," J Bus Res, vol. 125, pp. 120–134, Mar. 2021, doi: 10.1016/j.jbusres.2020.11.028.

[56] H. K. S. Lam, A. C. L. Yeung, and T. C. E. Cheng, "The impact of firms' social media initiatives on operational efficiency and innovativeness," Journal of Operations Management, vol. 47–48, no. 1, pp. 28–43, Nov. 2016, doi: 10.1016/j.jom.2016.06.001.

[57] C. Archer-Brown and J. Kietzmann, "Strategic knowledge management and enterprise social media," Journal of Knowledge Management, vol. 22, no. 6, pp. 1288–1309, Jul. 2018, doi: 10.1108/JKM-08-2017-0359.

[58] A. Bilgihan, A. Barreda, F. Okumus, and K. Nusair, "Consumer perception of knowledge-sharing in travel-related online social networks," Tour Manag, vol. 52, pp. 287–296, Feb. 2016, doi: 10.1016/j.tourman.2015.07.002.

[59] D. Schlagwein and M. Hu, "How and why organisations use social media: Five use types and their relation to absorptive capacity," Journal of Information Technology, vol. 32, no. 2, pp. 194–209, Jun. 2017, doi: 10.1057/jit.2016.7.

[60] S. Wang and R. A. Noe, "Knowledge sharing: A review and directions for future research," Human Resource Management Review, vol. 20, no. 2, pp. 115–131, Jun. 2010, doi: 10.1016/j.hrmr.2009.10.001.

[61] S. Panahi, J. Watson, and H. Partridge, "Information encountering on social media and tacit knowledge sharing," J Inf Sci, vol. 42, no. 4, pp. 539–550, Aug. 2016, doi: 10.1177/0165551515598883.

[62] M. S. Aslam, I. Ali, A. Qammar, L. Kiwan, and A. Dhir, "How knowledge acquisition creates a competitive edge? A qualitative inquiry from international consultancy alliance,"
International Marketing Review, vol. 39, no. 3, pp. 653–681, Jun. 2022, doi: 10.1108/IMR-03-2021-0111.

[63] Y. Zhang, M. Zhang, N. Luo, Y. Wang, and T. Niu, "Understanding the formation mechanism of high-quality knowledge in social question and answer communities: A knowledge co-creation perspective," Int J Inf Manage, vol. 48, pp. 72–84, Oct. 2019, doi:

10.1016/j.ijinfomgt.2019.01.022.

[64] Irfan, I., Sumbal, M. S. U. K., Khurshid, F., & Chan, F. T. (2022). Toward a resilient supply chain model: critical role of knowledge management and dynamic capabilities. Industrial Management & Data Systems, (ahead-of-print). [65] B. Kanter and K. D. Paine, "Go Beyond Counting Tweets to Find Success With Social Media," The Chronicle of Philanthropy, 2012. [Online]. Available:

https://www.philanthropy.com/article/go-beyond-counting-tweets-to-find-success-with-socialmedia/

[66] Dwivedi, Y. K., Kshetri, N., Hughes, L., Slade, E. L., Jeyaraj, A., Kar, A. K., ... & Wright, R. (2023). "So what if ChatGPT wrote it?" Multidisciplinary perspectives on opportunities, challenges and implications of generative conversational AI for research, practice and policy. International Journal of Information Management, 71, 102642.

[67] L. Bizzi, "Employees who use social media for work are more engaged—but also more likely to leave their jobs," Harvard Business Review, 2018. [Online]. Available:

https://hbr.org/2018/05/employees-who-use-social-media-for-work-are-more-engaged-but-alsomore-likely-to-leave-their-jobs

[68] K. K. Kapoor, K. Tamilmani, N. P. Rana, P. Patil, Y. K. Dwivedi, and S. Nerur,
"Advances in Social media research: Past, present and future," Information Systems Frontiers,
vol. 20, no. 3, pp. 531–558, Jun. 2018, doi: 10.1007/s10796-017-9810-y.

[69] X. Yang, H. (Jonathan) Ye, and X. Wang, "Social media use and work efficiency:
Insights from the theory of communication visibility," Information & Management, vol. 58, no.
4, p. 103462, Jun. 2021, doi: 10.1016/j.im.2021.103462.

[70] Felin, T., Foss, N. J., & Ployhart, R. E. (2015). The microfoundations movement in strategy and organization theory. Academy of Management Annals, 9(1), 575-632.

[71] Cowen, A. P., Rink, F., Cuypers, I. R., Grégoire, D. A., & Weller, I. (2022). Applying Coleman's boat in management research: Opportunities and challenges in bridging macro and micro theory. Academy of Management Journal, 65(1), 1-10. [72] Ferraris, A., Degbey, W. Y., Singh, S. K., Bresciani, S., Castellano, S., Fiano, F., & Couturier, J. (2022). Microfoundations of strategic agility in emerging markets: empirical evidence of Italian MNEs in India. Journal of World Business, 57(2), 101272.

[73] Palmié, M., Rüegger, S., & Parida, V. (2023). Microfoundations in the strategic management of technology and innovation: Definitions, systematic literature review, integrative framework, and research agenda. Journal of Business Research, 154, 113351.

[74] D. M. Oppenheimer and E. Kelso, "Information processing as a paradigm for decision making," Annu Rev Psychol, vol. 66, no. 1, pp. 277–294, Jan. 2015, doi: 10.1146/annurev-psych-010814-015148.

[75] R. Barkemeyer, C. Faugère, O. Gergaud, and L. Preuss, "Media attention to large-scale corporate scandals: Hype and boredom in the age of social media," J Bus Res, vol. 109, pp. 385–398, Mar. 2020, doi: 10.1016/j.jbusres.2019.12.011.

[76] K.-Y. Kwahk and D.-H. Park, "Leveraging your knowledge to my performance: The impact of transactive memory capability on job performance in a social media environment," Comput Human Behav, vol. 80, pp. 314–330, Mar. 2018, doi: 10.1016/j.chb.2017.10.047.

[77] T. Schubert, "Interventions to influence causal reasoning," Nature Reviews Psychology, vol. 1, no. 3, pp. 131–131, Mar. 2022, doi: 10.1038/s44159-022-00032-4.

[78] V. L. Mitchell and R. W. Zmud, "the effects of coupling it and work process strategies in redesign projects," Organization Science, vol. 10, no. 4, pp. 424–438, Aug. 1999, doi: 10.1287/orsc.10.4.424.

[79] A. Cordella and C. M. Bonina, "A public value perspective for ICT enabled public sector reforms: A theoretical reflection," Gov Inf Q, vol. 29, no. 4, pp. 512–520, Oct. 2012, doi: 10.1016/j.giq.2012.03.004.

[80] A. Lovari and C. Valentini, "Public sector communication and social media," in The Handbook of Public Sector Communication, Wiley, 2020, pp. 315–328. doi: 10.1002/9781119263203.ch21.

[81] D. E. Leidner, E. Gonzalez, and H. Koch, "An affordance perspective of enterprise social media and organizational socialization," in Strategic Information Management, 5th Edition., Routledge, 2020, pp. 364–402.

[82] Y. Sun et al., "Dark side of enterprise social media usage: A literature review from the conflict-based perspective," Int J Inf Manage, vol. 61, p. 102393, Dec. 2021, doi: 10.1016/j.ijinfomgt.2021.102393.

[83] Chaudhary, S., Dhir, A., Gligor, D., Khan, S. J., & Ferraris, A. (2022). Paradoxes and coping mechanisms in the servitisation journey. Industrial Marketing Management, 106, 323-337.

[84] P. M. Leonardi and E. Vaast, "Social Media and their affordances for organizing: A review and agenda for research," Academy of Management Annals, vol. 11, no. 1, pp. 150–188, Jan. 2017, doi: 10.5465/annals.2015.0144.

[85] S. Heinecke, "The game of trust: Reflections on truth and trust in a shifting media
 ecosystem," in Media Trust in a Digital World, Cham: Springer International Publishing, 2019,
 pp. 3–13. doi: 10.1007/978-3-030-30774-5_1.

[86] J. L. Pierce and A. L. Delbecq, "Organization Structure, Individual Attitudes and Innovation," The Academy of Management Review, vol. 2, no. 1, p. 27, Jan. 1977, doi: 10.2307/257602.

[87] S. Solaimani, T. van Eck, H. Kievit, and K. Koelemeijer, "An exploration of the applicability of Lean Startup in small non-digital firms: an effectuation perspective,"

International Journal of Entrepreneurial Behavior & Research, vol. 28, no. 9, pp. 198–218, Jul. 2022, doi: 10.1108/IJEBR-04-2021-0270.

[88] J. H. Grimm, J. S. Hofstetter, and J. Sarkis, "Corporate sustainability standards in multitier supply chains – an institutional entrepreneurship perspective," Int J Prod Res, pp. 1–23, Jan. 2022, doi: 10.1080/00207543.2021.2017053.

[89] S. Kurpjuweit, D. Reinerth, and S. M. Wagner, "Supplier innovation push," Research-Technology Management, vol. 61, no. 2, pp. 47–55, Mar. 2018, doi:

10.1080/08956308.2018.1399025.

[90] R. Strange and A. Zucchella, "Industry 4.0, global value chains and international business," Multinational Business Review, vol. 25, no. 3, pp. 174–184, Sep. 2017, doi: 10.1108/MBR-05-2017-0028.

[91] R. Hassan, R. de Neufville, and D. McKinnon, "Value-at-risk analysis for real options in complex engineered systems," in 2005 IEEE International Conference on Systems, Man and Cybernetics, pp. 3697–3704. doi: 10.1109/ICSMC.2005.1571721.

[92] K. E. K. Möller and P. Törrönen, "Business suppliers' value creation potential,"
Industrial Marketing Management, vol. 32, no. 2, pp. 109–118, Feb. 2003, doi: 10.1016/S0019-8501(02)00225-0.

[93] S. Mangla, J. Madaan, and F. T. S. Chan, "Analysis of flexible decision strategies for sustainability-focused green product recovery system," Int J Prod Res, vol. 51, no. 11, pp. 3428–3442, Jun. 2013, doi: 10.1080/00207543.2013.774493.

[94] Wan, J., Li, X., Dai, H. N., Kusiak, A., Martinez-Garcia, M., & Li, D. (2020). Artificialintelligence-driven customized manufacturing factory: key technologies, applications, and challenges. Proceedings of the IEEE, 109(4), 377-398. [95] P. C. Nutt, "Flexible decision styles and the choices of top executives," Journal of Management Studies, vol. 30, no. 5, pp. 695–721, Sep. 1993, doi: 10.1111/j.1467-6486.1993.tb00322.x.

[96] L. Cui, H. Wu, and J. Dai, "Modelling flexible decisions about sustainable supplier selection in multitier sustainable supply chain management," Int J Prod Res, pp. 1–22, May 2021, doi: 10.1080/00207543.2021.1924412.

[97] N. Khakzad, F. Khan, and P. Amyotte, "Dynamic safety analysis of process systems by mapping bow-tie into Bayesian network," Process Safety and Environmental Protection, vol. 91, no. 1–2, pp. 46–53, Jan. 2013, doi: 10.1016/j.psep.2012.01.005.

[98] E. Ambrose, D. Marshall, and D. Lynch, "Buyer supplier perspectives on supply chain relationships," International Journal of Operations & Production Management, vol. 30, no. 12, pp. 1269–1290, Nov. 2010, doi: 10.1108/01443571011094262.

[99] T. Schoenherr and M. Swink, "Revisiting the arcs of integration: Cross-validations and extensions," Journal of Operations Management, vol. 30, no. 1–2, pp. 99–115, Jan. 2012, doi: 10.1016/j.jom.2011.09.001.

[100] V.-H. Lee, K.-B. Ooi, A. Y.-L. Chong, and C. Seow, "Creating technological innovation via green supply chain management: An empirical analysis," Expert Syst Appl, vol. 41, no. 16, pp. 6983–6994, Nov. 2014, doi: 10.1016/j.eswa.2014.05.022.

[101] H. Lee, S.-B. Yang, and C. Koo, "Exploring the effect of Airbnb hosts' attachment and psychological ownership in the sharing economy," Tour Manag, vol. 70, pp. 284–294, Feb. 2019, doi: 10.1016/j.tourman.2018.08.017.

[102] H. Min, The essentials of supply chain management: New business concepts and applications. Pearson, 2015.

[103] M. P. Bhandarkar and R. Nagi, "STEP-based feature extraction from STEP geometry for Agile Manufacturing," Comput Ind, vol. 41, no. 1, pp. 3–24, Jan. 2000, doi: 10.1016/S0166-3615(99)00040-8.

[104] M. Paasivaara, B. Behm, C. Lassenius, and M. Hallikainen, "Large-scale agile transformation at Ericsson: a case study," Empir Softw Eng, vol. 23, no. 5, pp. 2550–2596, Oct. 2018, doi: 10.1007/s10664-017-9555-8.

[105] G. Serafeim, A. M. Rischbieth, and H. K. Koh, "Sustainability, business, and health,"JAMA, vol. 324, no. 2, p. 147, Jul. 2020, doi: 10.1001/jama.2020.8714.

[106] N. Farmer, The invisible organization: How informal networks can lead organizational change. Routledge, 2017. doi: 10.4324/9781315238883.

[107] Bouguerra, A., Gölgeci, İ., Gligor, D. M., & Tatoglu, E. (2019). How do agile organizations contribute to environmental collaboration? Evidence from MNEs in Turkey. Journal of International Management, 10071.

[108] J. Sarkis, "Supply chain sustainability: learning from the COVID-19 pandemic,"
International Journal of Operations & Production Management, vol. 41, no. 1, pp. 63–73, Dec.
2020, doi: 10.1108/IJOPM-08-2020-0568.

[109] M. Yang and P. Gabrielsson, "Entrepreneurial marketing of international high-tech business-to-business new ventures: A decision-making process perspective," Industrial Marketing Management, vol. 64, pp. 147–160, Jul. 2017, doi: 10.1016/j.indmarman.2017.01.007.
[110] Kazmaier, J., & van Vuuren, J. H. (2020). A generic framework for sentiment analysis: Leveraging opinion-bearing data to inform decision making. Decision Support Systems, 135, 113304.

[111] J. Winnard, A. Adcroft, J. Lee, and D. Skipp, "Surviving or flourishing? Integrating business resilience and sustainability," Journal of Strategy and Management, vol. 7, no. 3, pp. 303–315, Aug. 2014, doi: 10.1108/JSMA-11-2012-0059.

[112] Giacosa, E., Culasso, F., & Crocco, E. (2022). Customer agility in the modern automotive sector: how lead management shapes agile digital companies. Technol Forecast Soc Change, 175, 121362.

[113] F.-J. Braña, "A fourth industrial revolution? Digital transformation, labor and work organization: a view from Spain," Journal of Industrial and Business Economics, vol. 46, no. 3, pp. 415–430, Sep. 2019, doi: 10.1007/s40812-019-00122-0.

[114] T. M. Nisar, G. Prabhakar, and L. Strakova, "Social media information benefits, knowledge management and smart organizations," J Bus Res, vol. 94, pp. 264–272, Jan. 2019, doi: 10.1016/j.jbusres.2018.05.005.

[115] J. Hall, "Environmental supply chain innovation," in Greening the Supply Chain, Springer London, 2006, pp. 233–249. doi: 10.1007/1-84628-299-3_13.

[116] Hahn, G. J. (2020). Industry 4.0: a supply chain innovation perspective. International Journal of Production Research, 58(5), 1425-1441.

[117] Kirchoff, J. F., & Falasca, M. (2022). Environmental differentiation from a supply chain practice view perspective. International Journal of Production Economics, 244, 108365.

[118] Loonam, J., & O'Regan, N. (2022). Global value chains and digital platforms:Implications for strategy. Strategic Change, 31(1), 161-177.

[119] Ahammad, M. F., Basu, S., Munjal, S., Clegg, J., & Shoham, O. B. (2021). Strategic agility, environmental uncertainties and international performance: The perspective of Indian firms. Journal of World Business, 56(4), 101218.

[120] Christofi, M., Pereira, V., Vrontis, D., Tarba, S., & Thrassou, A. (2021). Agility and flexibility in international business research: A comprehensive review and future research directions. Journal of World Business, 56(3), 101194.

[121] Huang, P. Y., Pan, S. L., & Ouyang, T. H. (2014). Developing information processing capability for operational agility: implications from a Chinese manufacturer. European Journal of Information Systems, 23, 462-480.

[122] Clauss, T., Abebe, M., Tangpong, C., & Hock, M. (2019). Strategic agility, business model innovation, and firm performance: an empirical investigation. IEEE Trans Eng Manag, 68(3), 767-784.

[123] Department of Statistics Malaysia (DOSM), "Malaysia economic performance 2019,"2020.

https://www.dosm.gov.my/v1/index.php?r=column/cthemeByCat&cat=153&bul_id=bVN1K0tx TSt1TVRGRFZBRE8yU0JYZz09&menu_id=TE5CRUZCblh4ZTZMODZIbmk2aWRRQT09

[124] A. K. H. Yap and S. T. Yap, "COSO enterprise risk management: small-medium enterprises evidence," Asia-Pacific Management Accounting Journal, vol. 13, no. 2, pp. 83–111, 2018.

[125] N. Saputra, S. B. Abdinagoro, and E. A. Kuncoro, "The mediating role of learning agility on the relationship between work engagement and learning culture," Pertanika Journal of Social Sciences & Humanities, vol. 16, no. T, pp. 117–130, 2018, [Online]. Available: http://www.pertanika.upm.edu.my/resources/files/Pertanika%20PAPERS/JSSH%20Vol.%2026

%20(T)%20Mar.%202018/11%20JSSH(T)-0669-2018-6thProof.pdf

[126] J. S. Armstrong and T. S. Overton, "Estimating nonresponse bias in mail surveys,"
Journal of Marketing Research, vol. 14, no. 3, pp. 396–402, Aug. 1977, doi:
10.1177/002224377701400320.

[127] A. M. Ghouri and V. Mani, "Role of real-time information-sharing through SaaS: An industry 4.0 perspective," Int J Inf Manage, vol. 49, pp. 301–315, Dec. 2019, doi: 10.1016/j.ijinfomgt.2019.05.026.

[128] P. M. Podsakoff, S. B. MacKenzie, J.-Y. Lee, and N. P. Podsakoff, "Common method biases in behavioral research: A critical review of the literature and recommended remedies.,"
Journal of Applied Psychology, vol. 88, no. 5, pp. 879–903, 2003, doi: 10.1037/0021-9010.88.5.879.

[129] Change, S., Witteloostuijn, A. V., & Eden, L. (2010). From the editors: Common method variance in international research. Journal of International Business Studies, 41(2), 178.

[130] Doty, D. H., & Glick, W. H. (1998). Common methods bias: does common methods variance really bias results? Organizational Research Methods, 1(4), 374-406.

[131] A. M. Ghouri, V. Mani, M. R. Khan, N. R. Khan, and A. P. Srivastava, "Enhancing business performance through green human resource management practices: an empirical evidence from Malaysian manufacturing industry," International Journal of Productivity and Performance Management, vol. 69, no. 8, pp. 1585–1607, Aug. 2020, doi: 10.1108/IJPPM-11-2019-0520.

[132] Podsakoff, P. M., MacKenzie, S. B., & Podsakoff, N. P. (2012). Sources of method bias in social science research and recommendations on how to control it. Annual Review of Psychology, 63, 539-569. [133] MacKenzie, S. B., & Podsakoff, P. M. (2012). Common method bias in marketing:Causes, mechanisms, and procedural remedies. Journal of Retailing, 88(4), 542-555.

[134] E. W. K. Tsang, "Acquiring knowledge by foreign partners from international joint ventures in a transition economy: learning-by-doing and learning myopia," Strategic Management Journal, vol. 23, no. 9, pp. 835–854, Sep. 2002, doi: 10.1002/smj.251.

[135] B. Larrañeta, S. A. Zahra, and J. L. G. González, "Enriching strategic variety in new ventures through external knowledge," J Bus Ventur, vol. 27, no. 4, pp. 401–413, Jul. 2012, doi: 10.1016/j.jbusvent.2011.11.004.

[136] K. Z. Zhou and C. B. Li, "How knowledge affects radical innovation: Knowledge base, market knowledge acquisition, and internal knowledge sharing," Strategic Management Journal, vol. 33, no. 9, pp. 1090–1102, Sep. 2012, doi: 10.1002/smj.1959.

[137] D.-W. Kwak, Y.-J. Seo, and R. Mason, "Investigating the relationship between supply chain innovation, risk management capabilities and competitive advantage in global supply chains," International Journal of Operations & Production Management, vol. 38, no. 1, pp. 2–21, Jan. 2018, doi: 10.1108/IJOPM-06-2015-0390.

[138] S. M. Lee, D. Lee, and M. J. Schniederjans, "Supply chain innovation and organizational performance in the healthcare industry," International Journal of Operations & Production Management, vol. 31, no. 11, pp. 1193–1214, Oct. 2011, doi: 10.1108/01443571111178493.
[139] Y.-J. Seo, J. Dinwoodie, and D.-W. Kwak, "The impact of innovativeness on supply

chain performance: is supply chain integration a missing link?," Supply Chain Management: An International Journal, vol. 19, no. 5/6, pp. 733–746, Sep. 2014, doi: 10.1108/SCM-02-2014-0058.

[140] J. B. Bernerth and H. Aguinis, "A critical review and best-practice recommendations for control variable usage," Pers Psychol, vol. 69, no. 1, pp. 229–283, Feb. 2016, doi: 10.1111/peps.12103.

[141] T. Jambulingam and J. R. Nevin, "Influence of franchisee selection criteria on outcomes desired by the franchisor," J Bus Ventur, vol. 14, no. 4, pp. 363–395, Jul. 1999, doi: 10.1016/S0883-9026(98)00023-8.

[142] H. Jiao, C. K. Koo, and Y. Cui, "Legal environment, government effectiveness and firms' innovation in China: Examining the moderating influence of government ownership," Technol Forecast Soc Change, vol. 96, pp. 15–24, Jul. 2015, doi: 10.1016/j.techfore.2015.01.008.

[143] N. Wilson, M. Wright, and A. Altanlar, "The survival of newly-incorporated companies and founding director characteristics," International Small Business Journal: Researching Entrepreneurship, vol. 32, no. 7, pp. 733–758, Nov. 2014, doi: 10.1177/0266242613476317.

[145] T. C. Flatten, G. I. Greve, and M. Brettel, "Absorptive capacity and firm performance in SMEs: The mediating influence of strategic alliances," European Management Review, vol. 8, no. 3, pp. 137–152, Sep. 2011, doi: 10.1111/j.1740-4762.2011.01015.x.

[146] S. A. Zahra, H. J. Sapienza, and P. Davidsson, "Entrepreneurship and dynamic capabilities: A review, model and research agenda*," Journal of Management Studies, vol. 43, no. 4, pp. 917–955, Jun. 2006, doi: 10.1111/j.1467-6486.2006.00616.x.

[147] K. D. Brouthers, L. E. Brouthers, and S. Werner, "Influences on strategic decision-making in the Dutch financial services industry," J Manage, vol. 26, no. 5, pp. 863–883, Oct. 2000, doi: 10.1177/014920630002600506.

[148] B. M. Ruf, K. Muralidhar, R. M. Brown, J. J. Janney, and K. Paul, "An empirical investigation of the relationship between change in corporate social performance and financial

performance: A stakeholder theory perspective," Journal of Business Ethics, vol. 32, no. 2, pp. 143–156, 2001, doi: 10.1023/A:1010786912118.

[149] X. Zhai and Y. An, "Analyzing influencing factors of green transformation in China's manufacturing industry under environmental regulation: A structural equation model," J Clean Prod, vol. 251, p. 119760, Apr. 2020, doi: 10.1016/j.jclepro.2019.119760.

[150] S. B. Banerjee, E. S. Iyer, and R. K. Kashyap, "Corporate environmentalism:

Antecedents and influence of industry type," J Mark, vol. 67, no. 2, pp. 106–122, Apr. 2003, doi: 10.1509/jmkg.67.2.106.18604.

[151] M. A. Hitt, R. D. Ireland, and G. Stadter, "Functional importance and company performance: Moderating effects of grand strategy and industry type," Strategic Management Journal, vol. 3, no. 4, pp. 315–330, Oct. 1982, doi: 10.1002/smj.4250030404.

[152] R. Marais, S. Grobbelaar, and I. de Kock, "Healthcare technology transfer in Sub-Saharan Africa: Conceptual framework evaluation," IEEE Trans Eng Manag, pp. 1–17, 2022, doi: 10.1109/TEM.2021.3109445.

[153] T. A. Brown, Confirmatory factor analysis for applied research, Second Edition. Guilford publications, 2015.

[154] A. Boomsma, "Reporting analyses of covariance structures," Struct Equ Modeling, vol.
7, no. 3, pp. 461–483, Jul. 2000, doi: 10.1207/S15328007SEM0703_6.

[155] Y. Xia and Y. Yang, "RMSEA, CFI, and TLI in structural equation modeling with ordered categorical data: The story they tell depends on the estimation methods," Behav Res Methods, vol. 51, no. 1, pp. 409–428, Feb. 2019, doi: 10.3758/s13428-018-1055-2.

[156] M. Shevlin and J. N. V. Miles, "Effects of sample size, model specification and factor loadings on the GFI in confirmatory factor analysis," Pers Individ Dif, vol. 25, no. 1, pp. 85–90, Jul. 1998, doi: 10.1016/S0191-8869(98)00055-5.

[157] C. Fornell and D. F. Larcker, "Structural equation models with unobservable variables and measurement error: Algebra and statistics," Journal of Marketing Research, vol. 18, no. 3, p. 382, Aug. 1981, doi: 10.2307/3150980.

[158] Akhtar, P., Ghouri, A. M., Saha, M., Khan, M. R., Shamim, S., & Nallaluthan, K. (2022). Industrial digitization, the use of real-time information, and operational agility: Digital and information perspectives for supply chain resilience. IEEE Trans Eng Manag.

[159] J. F. Hair, R. E. Anderson, R. L. Tatham, and W. C. Black, Multivariate data analysis, 5th Edition. Upper Saddle River, NJ, United States: Prentice-Hall, Inc., 1998.

[160] B. M. Byrne, Structural equation modeling with AMOS. Routledge, 2013. doi: 10.4324/9780203805534.

[161] S. Chatterjee, R. Chaudhuri, D. Vrontis, and R. Piccolo, "Enterprise social network for knowledge sharing in MNCs: Examining the role of knowledge contributors and knowledge seekers for cross-country collaboration," Journal of International Management, vol. 27, no. 1, p. 100827, Mar. 2021, doi: 10.1016/j.intman.2021.100827.

[162] Ghouri, A. M., Akhtar, P., Haq, M. A., Mani, V., Arsenyan, G., & Meyer, M. (2021).
Real-time information sharing, customer orientation, and the exploration of intra-service industry differences: Malaysia as an emerging market. *Technological Forecasting and Social Change*, *167*, 120684.

[163] M. Ghasemaghaei, K. Hassanein, and O. Turel, "Increasing firm agility through the use of data analytics: The role of fit," Decis Support Syst, vol. 101, pp. 95–105, Sep. 2017, doi: 10.1016/j.dss.2017.06.004.

[164] S. Gupta, A. Leszkiewicz, V. Kumar, T. Bijmolt, and D. Potapov, "Digital analytics:
Modeling for Insights and new methods," Journal of Interactive Marketing, vol. 51, pp. 26–43,
Aug. 2020, doi: 10.1016/j.intmar.2020.04.003.

[165] S. Forkmann, S. C. Henneberg, and M. Mitrega, "Capabilities in business relationships and networks: Research recommendations and directions," Industrial Marketing Management, vol. 74, pp. 4–26, Oct. 2018, doi: 10.1016/j.indmarman.2018.07.007.

[166] R. Rajaguru, M. J. Matanda, and W. Zhang, "Supply chain finance in enhancing supplyoriented and demand-oriented performance capabilities – moderating role of perceived partner opportunism," Journal of Business & Industrial Marketing, Jan. 2022, doi: 10.1108/JBIM-11-2020-0487.

[167] S. Goh and G. Richards, "Benchmarking the learning capability of organizations,"
European Management Journal, vol. 15, no. 5, pp. 575–583, Oct. 1997, doi: 10.1016/S0263-2373(97)00036-4.

[168] H. Cao and P. Folan, "Product life cycle: the evolution of a paradigm and literature review from 1950–2009," Production Planning & Control, vol. 23, no. 8, pp. 641–662, Aug. 2012, doi: 10.1080/09537287.2011.577460.

[169] G. S. Day, "The product life cycle: Analysis and applications issues," J Mark, vol. 45, no.
4, pp. 60–67, Sep. 1981, doi: 10.1177/002224298104500408.

[170] T. D. Hedberg, N. W. Hartman, P. Rosche, and K. Fischer, "Identified research directions for using manufacturing knowledge earlier in the product life cycle," Int J Prod Res, vol. 55, no. 3, pp. 819–827, Feb. 2017, doi: 10.1080/00207543.2016.1213453.

[171] K. B. Munksgaard, J. Stentoft, and A. Paulraj, "Value-based supply chain innovation,"
Operations Management Research, vol. 7, no. 3–4, pp. 50–62, Dec. 2014, doi: 10.1007/s12063-014-0092-y.

[172] C. K. Prahalad and V. Ramaswamy, "Co-opting Customer Competence," Harvard business review, 2000. [Online]. Available: <u>https://hbr.org/2000/01/co-opting-customer-competence</u>

Appendix A:

Questionnaire item description.

Knowledge acquisition from social media:

- 1. Our company uses a system to continuously collect information from customers through social media.
- 2. Our company uses a system to continuously collect information about competitors' activities through social media.
- 3. Our company uses a system to continuously collect information from suppliers through social media.
- 4. Our company uses a system to continuously collect information from intermediaries through social media.
- 5. Our company uses a system to continuously collect information from governments through social media.

Organizational microfoundation structure and design:

- 1. Our company's employees utilize personal networks with our strategic partners.
- 2. Our company uses flexible decision-making structures to understand key factors.
- 3. Our company uses open communication channels with our partners and managers

Supply chain innovation:

- 1. We pursue a cutting-edge (leading technology) system that can integrate relevant information.
- 2. We consider modern transport systems (e.g., trucks, vans) for supply chain innovation.
- 3. We pursue continuous innovation in our core supply chain processes.
- 4. We pursue agile supply chain processes against market changes.
- 5. In our organization, creative methods are taken into account for supply chain operations.

Operational agility:

- 1. The reliability of our offerings has increased.
- 2. Our day-to-day operations are flexible enough to deal with customized demand.
- 3. Our offerings are more cost-efficient than those of our competitors.
- 4. We are very quick in delivering our offerings.
- 5. Our responses to market changes are very reliable.
- 6. We are very flexible in our offerings in order to adapt to market changes.
- 7. We promptly redesign our offerings to adapt to market changes.
- 8. We are very quick in exploiting market opportunities